Author's response to reviews

Title: Treatment of osteoporosis in an older home care population

Authors:

Shelly A Vik (savik@ucalgary.ca)
Colleen J Maxwell (maxwell@ucalgary.ca)
David A Hanley (dahanley@ucalgary.ca)

Version: 4 Date: 28 January 2005

Author’s response to reviews:

Dear Editorial Team:

The requested formatting changes have been completed:
- References are now placed after full stops.
- The 'Figure' has been reformatted according to the specifications indicated on link you provided.

We thank you for your patience and support during the preparation of this manuscript.

Sincerely,
Shelly A. Vik
Colleen J. Maxwell
David A. Hanley